Dear reviewer, thank you very much for your positive comment and recommendation to accept the manuscript. A language revision has been carried out.

Dear Reviewer:

Thank you very much for your comments to improve our manuscript. All your comments have been taken into account: a revision of the language has been carried out, specific information has been added to the recommendations for clinical practice in the core tip. In addition, a table with a decision making diagram has been added to indicate which type of treatment should be carried out based on the patient's assessment. Finally, the conclusion has been improved and is now much more robust and gives an idea of how the patient should be approached. Thank you very much for everything. Regards